Author's response to reviews

Title: Weight Watchers on prescription: An observational study of weight change among adults referred to Weight Watchers by the NHS

Authors:

Amy L Ahern (Amy.Ahern@mrc-hnr.cam.ac.uk)
Ashley D Olson (Ashley.Olson@mrc-hnr.cam.ac.uk)
Louise M Aston (louise.aston@cantab.net)
Susan A Jebb (Susan.Jebb@mrc-hnr.cam.ac.uk)

Version: 4 Date: 23 May 2011

Author's response to reviews:

Dear Editor,

We are very pleased with your recent decision to publish our article in BMC Public Health. Please find attached a revised manuscript which make the required formatting changes. Specific changes are listed below:

We now specifically state that all authors read and approved the final manuscript.

We have now checked the reference formatting to ensure references are cited in numerical order.

We have moved the legend in figure three from the manuscript to the figure.

We have removed the shading in Table 1 and made the text in this section bold.

We have removed all highlighting and tracking from the manuscript.

We have split the merged cells in table 1

We have cropped all the figures

We hope that this covers all changes requested and look forward to seeing the final manuscript in print.

Many thanks

Susan Jebb